246' Copyright Ron Burback 1997Groups, Profiles, and ObjectsRon BurbackLecture 12

247' Copyright Ron Burback 1997Application ServernOne of the jobs of the CDS is to provide the binding information for applications. nAn application entry in the CDS is an object that contains the interface identifier and the IP of the machine where the service is currently running.nThis information is placed their by the service.

248' Copyright Ron Burback 1997Application ClientnThe application client asks the name space for a service. The name space returns this binding.nThis is done in two stepsnget the IP from the cdsnget the port from the end point mapper at that IP

249' Copyright Ron Burback 1997CachednThis binding is then cached on the local machine by the clerk and eventually flushed to disk. When the application client wants the information again, it comes from the cache. If the CDS is done, the information is used from the cache. nOnce the binding is established, the client talks directly to the server.

250' Copyright Ron Burback 1997Stale InformationnThe CDS is abundant with stale information.nThe Clerk caches are abundant with stale information.nThe client can force a cache re-fresh and go directly to the CDS for the best information.nThe rpc does auto time out and binding repair.

251' Copyright Ron Burback 1997Groups of ServicesnA group of services may be available. Any one service in the group can do the job. Here you have many identical services to handle the client request. nThis is done with a CDS group. Not the same as a Security group.

252' Copyright Ron Burback 1997Groupsnexample name or group /.:/mathnThere may be many math servers. Their names could be /.:/math/one, /.:/math/two, or /.:/math/three.nThe client asks for math and will randomly get the bindings for one, two, or three. nIf the service goes off line, the rpc will go find another service.

253' Copyright Ron Burback 1997Why Groups?nLoad balancingnChange in the load requirements. If you need more machine, no problem. If you need less machines, no problem.nFault tolerance. If the services fails, no problems, just try another service. The rpc plus a few lines of code, will do this for you. nParallelism of tasks.

254' Copyright Ron Burback 1997ProfilesnProfiles are just like groups except they have a priority for each entry. You just don�t get a service but the best service based on the priority.

255' Copyright Ron Burback 1997The profile hierarchynThe profiles form a hierarchy. nthe root of the hierarchy is /.:/cell-profile.nThe system will try to find servers, following the hierarchy, until a binding is established.

256' Copyright Ron Burback 1997What is a binding?nIPnVersion numbernUUID of the interfacenPort numbernThere is a binding for each network protocol, tcp, udp

257' Copyright Ron Burback 1997End Point MappernThe end point mapper is a list of all services running on that machine. When a service starts, it registers with the end point mapper. nThe mapper runs on port 135